Author’s response to reviews

Title: Sero-Prevalence of Hepatitis B Virus and Associated Factors among Pregnant Women in Gambella Hospital, South Western Ethiopia: Facility based cross-sectional study

Authors:

Abayneh tanga (abaynehtun@yahoo.com)

Misanew Teshome (misanew08@gmail.com)

Desta Hiko (destahika@gmail.com)

Chaltu Fikru (fikiruc@yahoo.com)

Gemechu Jilo (gemechukejela86@gmail.com)

Version: 4 Date: 12 Mar 2019

Author’s response to reviews:

Reviewer 2:

comments on abstract section to correct Hepatitis b in to Hepatitis B

Authors: thank you, corrected accordingly

Reviewer 2: under ethical consideration to make small letter c

Authors: thank you, corrected accordingly

Reviewer 2: on result section, to make sentence complete add was

Authors: thank you so much, corrected accordingly

Reviewer 2: on result section, table number 1, to correct Characteristics in the table title in to characteristics and also to make table breaks, to write full and meaningful result finding like 0.72 instead of .72.

Authors: thank you so much for interest looking and amended accordingly.

Reviewer 2: on result section, table number 2, to make space between numbers and brackets, make table breaks, write full and meaningful result finding like 0.77, and also table consistency

Authors: thank you, corrected accordingly

Reviewer 2: on result section, table number 3, to make table consistent
Reviewer 2: on abbreviation, write in sentence look

Authors: thank you, corrected accordingly

Regarding editors comment on authors specific contribution especially GK

Thank you for nice reminding! we have amended as,

GK: Involved in generating the concept of this research paper, proposal writing, designing, analysis, write-up, preparation of scientific paper, and manuscript preparation.